Author's response to reviews

Title: Cerebrospinal fluid HIV infection and pleocytosis: Relation to systemic infection and antiretroviral treatment

Authors:

Serena S Spudich (sspudich@sfgh.ucsf.edu)
Annelie C Nilsson (anilson@itsa.ucsf.edu)
Nicole D Lollo (nlollo@sfgh.ucsf.edu)
Teri J Liegler (tligler@gladstone.ucsf.edu)
Christos J Petropoulos (cpetropoulos@ViroLogic.com)
Steven G Deeks (sdeeks@php.ucsf.edu)
Ellen E Paxinos (epaxinos@ViroLogic.com)
Richard W Price (rprice@sfgh.ucsf.edu)

Version: 8 Date: 29 October 2005

Author's response to reviews:

The Editors,
BioMed Central Infectious Diseases October 27, 2005

Dear Editors,

We are delighted that our manuscript entitled "Cerebrospinal fluid HIV infection and pleocytosis: Relation to systemic infection and antiretroviral treatment" has been approved for publication in BMC Infectious Diseases. We have received the instructions for formatting changes to our manuscript and have made all of the indicated changes. In addition, we have proofread the manuscript again for other editorial issues (making no substantial changes). The reformatted manuscript and accompanying files are uploaded in conjunction with this cover letter.

Thank you very much. Please contact me with any further questions.

Sincerely,

Serena Spudich, MD
Assistant Adjunct Professor
Neurology, UCSF
San Francisco General Hospital, Rm 4M62
1001 Potrero Avenue
San Francisco, CA 94110 USA
tel (415) 206-4487
fax (415) 476-5582
email: sspudich@sfgh.ucsf.edu